Author’s response to reviews

Title: Comparative study of Interleukin-18 (IL-18) Serum levels in Adult Onset Still's Disease (AOSD) and Systemic Onset Juvenile Idiopathic Arthritis (sJIA) and its use as a Biomarker for Diagnosis and Evaluation of Disease Activity.

Authors:

Holger Kudela (holger.kudela@freenet.de)

Susanne Drynda (susanne.drynda@med.ovgu.de)

Anke Lux (anke.lux@med.ovgu.de)

Gerd Horneff (g.horneff@asklepios.com)

Jörn Kekow (joern.kekow@med.ovgu.de)

Version: 2 Date: 24 Sep 2018

Author’s response to reviews:

Dear editors,

Thank you for accepting our manuscript. I updated the references as requested.

Yours

Holger Kudela